Author's response to reviews

Title: Efficiency of immediate postoperative inpatient physical therapy following total knee arthroplasty: an RCT

Authors:

Anton F Lenssen (alen@pmzl.azm.nl)
Yvonne H.F Crijns (Ycrij@pmzl.azm.nl)
Eddie M.H. Waltje (ewal@pmzl.azm.nl)
Mike J.A. van Steyn (Mst@sort.azm.nl)
Ruud G.T. Geesink (RGE@sort.azm.nl)
Piet A van den Brandt (pa.vandenBrandt@epid.unimaas.nl)
Rob A de Bie (ra.debie@epid.unimaas.nl)

Version: 4 Date: 29 August 2006

Author's response to reviews:

Dear BioMed Central Editorial Production Team,
although we used the Biomedcentral link in our Endnote programm we seem to have made some mistakes in our reference list. We hope we changed the references according to your requests and double checked Medline abbreviations. I hope our new submission will be correct,
sorry for the inconvenience

Ton lenssen